Accelerator Status Report
Run Schedule April 23, 2009 to June 14, 2009

· Injector energy 66.40 MeV

· Linac Setup 590.20 MeV

· Pass setup: 1/5/5
· Wien angle: -14 degrees

· The machine was successfully restored at 5.9 GeV. Then it was scaled to 5.968 GeV.

· Optics is good and stable

· Injector power meter sometimes gets stuck. It can be reset from the inj service building.

· Q Measurement test plan running parasitically.
· 1L06 insulating vacuum pump was replaced due to a bad power cord.

· RF issues.

· ARC4 box supply has a DC overload fault. 

Hall A
E05-015/E08-005    1.246GeV     1 Pass

· Hall A beam was fine.
· Hall A SmartKnob: primary and secondary quads were re-labeled.
Hall B
Eg1-DVCS    5.968 GeV    5 Pass

· Hall B nA BPMs have problems indicating beam. 
· The up stream halo counts will be FSD interlocked.
Hall C

· The hall is not scheduled to run.
